

Author's response to reviews

Title: HVC1 ameliorates hyperlipidemia and inflammation in LDLR^{-/-} mice

Authors:

Se-Yun Cheon (chunsay1008@naver.com)

Kyung-Sook Chung (adella76@hanmail.net)

Kyung-Jin Lee (dostudy@naver.com)

Ho-Young Choi (hychoi@khu.ac.kr)

In-Hye Ham (sayxhd@naver.com)

Dong-Hoon Jung (brinicle87@naver.com)

Yun-Yeop Cha (omdcha@sangji.ac.kr)

Hyo-Jin An (hjan@sj.ac.kr)

Version: 1 Date: 30 Dec 2016

Author's response to reviews:

Dec-30-2016

Dear editor

BMC Complementary and Alternative Medicine

We appreciate reviewers' valuable comments and the opportunity to re-submit the manuscript, entitled "HVC1 ameliorates hyperlipidemia and inflammation in LDLR^{-/-} mice". Reviewers' comments are insightful and we have revised the manuscript accordingly. Below, we address point by point how we respond to reviewers' comments, and we hope that the revised manuscript is now adequate for publication in BMC Complementary and Alternative Medicine.